Author's response to reviews

Title: Factor Analysis of the Clustering of Common Somatic Symptoms: A Preliminary Study

Authors:

Chung-Huang Tsai (m333834@yahoo.com.tw)

Version: 4 Date: 3 June 2010

Author's response to reviews: see over
Dear editors
Thank you for your notify. I have revised the manuscript according the journal style.
Next I will reply them point-by-point.
Formatting changes to be made:
-----------------------------
Major revisions (we require the author to make these changes)
Affiliations: Each affiliation should be written in full, in the following format:
Department, Institute, (Street), City, (Postal code), Country. This may mean repeating
institutional addresses if the authors are from different departments within an
institution.
Reply
Chung-Huang Tsai (delete MD)

Affiliations have been hanged to request form.
Department of Family Medicine, Cheng Ching Hospital, No. 118 sec. 3 Chung-Kang
RD. Taichung 407, Taiwan
Graduate Institute of Biochemical Sciences and Technology, Chaoyang University of
Technology, No.168, Jifong E. Rd., Wufong Township, Taichung County 41349,
Taiwan
Department of Health Business Administration, HungKuang University, No 34,
Chung-Chie Rd, Sha Lu, Taichung, 443, Taiwan

Correspondence to:
Chung-Huang Tsai (delete MD)
Department of Family Medicine, Cheng Ching Hospital, No. 118 sec. 3 Chung-Kang
RD. Taichung 407, Taiwan

Manuscript: this manuscript does not have the correct structure. Please reformat it to
use these headings:

Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if
any); Competing interests; Authors' contributions; Authors? information (optional);
Acknowledgements (optional); References
Reply
I have revised the structure as needed according to above format.
Authors' contributions: I do it by myself, so I don’t declare it here.
Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Qualifications: Please remove the authors' qualifications (e.g. PhD / MD) or job titles from the manuscript and submission system.

Reply
I have deleted MD.

Title page: Please remove running title, and move Financial support statement to Acknowledgements section, just before the reference list at the end of the manuscript.

Reply
I have deleted running title, and moved Financial support statement to Acknowledgements section.

Key words: Please remove.

Reply
I have deleted Key words.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Reply
Some minor revisions were done.
Change men to males, women to females.
Change the bold format of male, female and $P$-value in Table 1.

Please contact us if you have any problems or questions.

Best wishes,

Chung-Huang Tsai